Author's response to reviews

Title: A draft framework for measuring progress towards the development of a national health information infrastructure

Authors:

Dean F. Sittig (dean.f.sittig@kp.org)
Richard N Shiffman (richard.shiffman@yale.edu)
Kevin Leonard (k.leonard@utoronto.ca)
Charles Friedman (friedmc1@mail.nlm.nih.gov)
Barbara Rudolph (brudolph@leapfroggroup.org)
George Hripcsak (hripcsak@columbia.edu)
Laura L Adams (ladams@riqi.org)
Lawrence C Kleinman (lkleinma@NEMOURS.ORG)
Rainu Kaushal (RKAUSHAL@PARTNERS.ORG)

Version: 6 Date: 1 June 2005

Author's response to reviews:

trying to get rid of old footnote that keeps popping in.
sorry.
dean